Author's response to reviews

Title: Systematic review and meta-analysis comparing land and aquatic exercise for people with hip or knee arthritis on function, mobility and other health outcomes

Authors:

Stephanie I Batterham (sibat2@student.monash.edu.au)
Sophie Heywood (physio@melbsportsmed.com.au)
Jennifer L Keating (jenny.keating@med.monash.edu.au)

Version: 9 Date: 21 May 2011

Author's response to reviews:

The following requests have been obliged

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Tables: please note that we are unable to display text vertically within the cells: please re-layout your table without these vertical elements. Also, please remove any visible vertical lines from the tables.

This is adjusted as requested

You have authors referenced by name in your tables, please also add the corresponding number from the reference list. All references in tables must also be in the reference list.

Adjusted as requested

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Please add the country detail to affiliation number 2.

Adjusted as requested

best wishes

Jenny